Check for updates

OPEN ACCESS

EDITED BY Mohammad Safiqul Islam, Noakhali Science and Technology University, Bangladesh

REVIEWED BY Jinsong Geng, Nantong University, China Md. Abdul Aziz, University of Manitoba, Canada

*CORRESPONDENCE Qijin Shu Shuqjhz@163.com

[†]These authors have contributed equally to this work and share first authorship

RECEIVED 15 July 2023 ACCEPTED 15 May 2024 PUBLISHED 03 June 2024

CITATION

Sun L, Zhao F, Xiang Y, Chen S and Shu Q (2024) Association of immune checkpoint inhibitors with SARS-CoV-2 infection rate and prognosis in patients with solid tumors: a systematic review and meta-analysis. *Front. Immunol.* 15:1259112. doi: 10.3389/fimmu.2024.1259112

COPYRIGHT

© 2024 Sun, Zhao, Xiang, Chen and Shu. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Association of immune checkpoint inhibitors with SARS-CoV-2 infection rate and prognosis in patients with solid tumors: a systematic review and meta-analysis

Lin Sun^{1†}, Fangmin Zhao^{1†}, Yuying Xiang¹, Shuyi Chen² and Qijin Shu^{2*}

¹The First School of Clinical Medicine, Zhejiang Chinese Medical University, Hangzhou, Zhejiang, China, ²Department of Oncology, The First Affiliated Hospital of Zhejiang Chinese Medical University (Zhejiang Provincial Hospital of Traditional Chinese Medicine), Hangzhou, Zhejiang, China

The rate and prognosis of severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) infection in patients with solid cancer tumors actively treated with immune checkpoint inhibitors (ICIs) have not been fully determined. The goal of this meta-analysis was to explore this issue, which can be helpful to clinicians in their decision-making concerning patient treatment. We conducted a thorough search for relevant cohort studies in the databases PubMed, Embase, Cochrane Library, and Web of Science. Mortality and infection rate were the primary endpoints, and the incidence of severe or critical disease was the secondary result. A total of 6,267 cases (individual patients) were represented in 15 studies. Prior exposure to ICIs was not correlated with an elevated risk of SARS-CoV-2 infection (relative risk (RR) 1.04, 95% CI 0.57-1.88, z = 0.12, P = 0.905) or mortality (RR 1.22, 95% CI 0.99–1.50, z = 1.90, P = 0.057). However, the results of the meta-analysis revealed that taking ICIs before SARS-CoV-2 diagnosis increased the chance of developing severe or critical disease (RR 1.51, 95% CI 1.09-2.10, z = 2.46, P = 0.014). No significant inter-study heterogeneity was observed. The infection and mortality rates of SARS-CoV-2 in patients with solid tumors who previously received ICIs or other antitumor therapies did not differ significantly. However, secondary outcomes showed that ICIs treatment before the diagnosis of SARS-CoV-2 infection was significantly associated with the probability of severe or critical illness.

Systematic review registration: https://www.crd.york.ac.uk/prospero/ #recordDetails PROSPERO, identifier CRD42023393511.

KEYWORDS

SARS-CoV-2, immune checkpoint inhibitors, cancer, infection, mortality, meta-analysis

1 Introduction

In late 2019 and early 2020 the outbreak of severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) rapidly spread to become a global pandemic and a serious public health emergency, and the associated syndrome was named coronavirus disease 2019 (COVID-19) (1). According to reports, patients with cancer are more likely to develop SARS-CoV-2 infection than healthy individuals, and this patient population is characterized by rapid disease progression, a higher proportion of individuals needing more intensive care, and a higher fatality rate (2-4). Therefore, there was a general trend at the beginning of the pandemic to stop cancer treatment because it was not known whether such treatments would affect the infection rate and prognosis of SARS-CoV-2. One of the most important cancer treatments in oncology is the use of immune checkpoint inhibitors (ICIs), a type of immunotherapy that improves patient outcomes in cancer cases. ICIs have shown significant efficacy in various cancers, such as lung cancer, melanoma, head and neck squamous cell carcinoma, and classical Hodgkin's lymphoma (5). The approved ICIs are monoclonal antibodies that block the proteins CTLA-4, PD-1, or PD-L1.

Due to tumor depletion, malnutrition, and decreased immune function caused by anti-tumor therapy, patients with malignant tumors are susceptible to SARS-CoV-2 infection. Therefore, most guidelines on the effects of ICIs on the course of COVID-19 have focused on their potentially harmful effects, especially on the development of the most serious inflammatory complications. Moreover, it is generally recommended to delay the use of ICIs in the SARS-CoV-2 environment. The National Comprehensive Cancer Network (NCCN) advises delaying antitumor treatment for patients who test positive for SARS-CoV-2; the precise amount of time depends on the severity of the infection (6), which can be a useful indicator in the absence of other factors. Other organizations have issued recommendations for treating patients with cancer. The German Society for Hematology and Medical Oncology issued a guideline that advocated delaying or stopping antineoplastic medicines during the current pandemic (7). The National Institute for Health and Care Excellence (NICE) advises that systemic anticancer treatment should ideally be delayed until the patient has been without severe symptoms for at least 10 days (8).

Because the majority of current guidelines recommend delaying ICI treatment, oncologists are becoming increasingly concerned about the safety of this treatment. Consequently, patients may have had their treatments postponed or permanently stopped, which could have had a significant impact on their prognosis (9–11). However, we note that there are many published studies in which the authors propose that ICI administration to patients with SARS-CoV-2 infection may be beneficial. According to Yekeduz et al. (12), the use of ICIs in patients with cancer during the pandemic proved safe. As reported by Pezeshki et al. (13), ICIs can be used as immunotherapy against SARS-CoV-2 in patients without cancer because they increase T-cell proliferation and activation. Moreover, ICIs stimulate T cells and may suppress viral infection (14, 15). ICIs have been reported to increase the absolute lymphocyte count in patients with cancer, which is considered a strong prognostic signal and a marker of therapy response (16).

We reviewed prior research, including the molecular processes of ICIs and the associated pathophysiological mechanisms of SARS-CoV-2 infection, to investigate the causes. Through the mediation of cellular immunological and cytotoxic capabilities, CD8+ T cells play a critical role in regulating pathogen infection (17) and are involved in viral clearance (18). The number of T lymphocytes has been shown to continuously decline in the peripheral blood of individuals with SARS-CoV-2-induced diseases (19). This decrease was more pronounced in CD8+ T cells after analyzing the cell counts and traits of various T cell subsets (20), and the low proportion of CD8+ T cells was associated with the occurrence of acute respiratory distress syndrome (ARDS) in patients with SARS-CoV-2 activity (21). However, during the development of the disease, a vicious circle is formed due to the presence of exhausted CD8+ T lymphocytes that frequently overexpress the inhibitory receptors CTLA4 and PD-1/PD-L1, resulting in impaired effector function and low proliferative potential (22).

Malignant tumors often express the PD-L1 and/or PD-L2 ligands, which bind to PD-1 on T cells and cause T cell "exhaustion." Thus, malignant cells can escape the anti-tumor defenses of the host. Through a variety of intricate activation mechanisms, ICIs cause T cells to activate, which kills cancer cells. The process begins with antigen presentation on the surface of antigen-presenting cells (APC) through the T cell receptor (TCR) and major histocompatibility complex (MHC) (23), and the costimulatory signal is controlled by the binding of B7-1 (CD80) or B7-2 (CD86) on the surface of APC to CD28 on the surface of T cells (24). Inhibitory T-cell receptors and their ligands include LAG3-MHC, CTLA4-B7, and PD1-PDL1. To restore the T-cellmediated antitumor immune response, anti-PD-1/PD-L1 antibodies act on the effector phase of the cancer immune cycle and block the PD-1/PD-L1 pathway (25). By inhibiting CTLA-4 from attaching to B7, the anti-CTLA-4 antibody improves the ability of CD28s to bind to B7, activates T cells, and has antitumor effects (26, 27). These results indicate that ICIs may reduce the infection rate or poor prognosis of SARS-CoV-2 by increasing the T-cell response to viruses (28).

Based on the clinical study results and the deduced mechanisms mentioned above, we propose that ICIs are safe and will not affect the SARS-CoV-2 infection rate or prognosis of patients with solid tumors. However, only a few meta-analyses have summarized the safety of ICIs administered to patients with cancer in the context of SARS-CoV-2 (29, 30), and none of them are specific to patients with solid tumors. Therefore, to explore the safety of ICIs treatment and provide evidence-based treatment insights and risk recommendations for the potential interaction between ICIs and COVID-19, we conducted this meta-analysis to compare the SARS-CoV-2 infection rate and prognosis in patients with solid tumors receiving ICIs (compared with those patients who did not receive ICIs treatment). we screen qualified studies with unified and scientific evaluation criteria, and apply statistical methods to evaluate the infection rate, mortality and severity of patients with solid tumor.

2 Material and methods

2.1 Search strategy

The Preferred Reporting Items for Systematic Reviews and Meta-analyses (PRISMA) guidelines (31) were followed in the design and reporting of our study, which has been registered in the PROSPERO database (CRD42023393511). A thorough search plan was then developed. Studies published from December 1, 2020, to January 29, 2023, were retrieved from the following databases: PubMed, Embase, the Cochrane Library, and Web of Science. The following keywords were included in our search, based on the PICOS principle: "Immune Checkpoint Inhibitor," "SARS-CoV-2," "Neoplasms," etc. The detailed search formulas are provided in Supplementary Material 1.

2.2 Criteria for inclusion and exclusion

The criteria for selecting studies were: (a) patients with solid cancer who received ICIs treatment were included in the study, (b) patients in the control group who received other types of antitumor therapies as comparators, and (c) SARS-CoV-2 infection and mortality were the primary outcomes, and severe/critical COVID-19 was the secondary outcome, and (d) cohort studies that computed RRs using pertinent statistics. In terms of the designation "severe" or "critical" illness, this was determined based on WHO guidelines (32, 33). We evaluated the criteria for severe respiratory disease, characterized by the need for an intensive care unit (ICU), or respiratory failure/acute respiratory distress syndrome (ARDS).

The following exclusion criteria were applied: (a) studies published only in abstract form, (b) studies in which patient data were missing for a group of patients, (c) studies involving patients with hematological malignancies, (d) studies in which data for patients receiving ICIs could not be distinguished from the data for the entire patient group, and (e) basic research, reviews, conferences, guidelines, editorials, comments, case reports, study protocols, and repeated publications.

2.3 Study selection

The screening procedure was performed by two independent authors who were blinded to one another. Full-text screening was performed to identify studies that met the inclusion and exclusion criteria. Disagreements between the reviewers were resolved through conversations with a third author.

2.4 Data extraction

One author extracted the data and created a specific table to house all study data, and another author double-checked the data extraction to ensure data accuracy, and the following details were recorded: name of the first author, publication year, country, study type, total sample size, proportion of males, median or mean age, tumor type, SARS-CoV-2 diagnostic method, ICIs interval before diagnosis of the virus, and study outcomes. The main outcomes were RRs with 95% CIs for infection rate and mortality, and the incidence of severe/critical disease was recorded as the secondary outcome. By extracting the total number of studies from the original information and the number of occurrences in the experimental and control groups, we calculated the incidence of events in each group and obtained the relative risk (RR) values. Specific information was obtained from published data or calculated using the original chart data.

2.5 Quality assessment

The Newcastle-Ottawa Scale (NOS) was used to rate the quality of the observational studies. Selection (4 points), comparability (2 points), and outcome (3 points) were the three elements of the scale (34, 35). If a study received at least seven out of nine overall responses, it was considered to be of excellent quality. Two reviewers evaluated the quality of the studies. A consensus was reached to address any discrepancies. The results of the quality assessment of the articles are summarized in Supplementary Material 2. The NOS ratings of the listed articles varied from 7 to 9, all of which were considered high quality.

2.6 Statistical analysis

Stata software (version 15.1) was used for all statistical analyses. Q and I² statistics (36, 37) were used to evaluate heterogeneity. Significant heterogeneity was indicated by $P \leq 0.05$ or I² $\geq 50\%$. Subgroup and meta-regression analyses were performed to investigate the potential causes of significant heterogeneity.

Currently, almost all commonly used meta-analysis software allows meta-analysis under two models: the fixed-effects model and the random-effects model. The commonly used selection principle is to select a model according to the level of heterogeneity. However, a few researchers believe this method does not conform to the inherent statistical hypotheses of the model (38). The fixed effects model assumes that all studies in the meta-analysis had the same effect size, whereas the random effects model assumes that different studies had different effect sizes. Although the two models are weighted by the inverse variance method, the fixed effects model only considers the sampling error; therefore, the weight of the large sample study was larger and that of the small sample study was smaller. In the randomeffects model, the decisiveness of sample size for weight was relatively weak. The differences in characteristics and sample sizes among the included studies are evident in the data discussed in Table 1, indicating the difficulty in maintaining a high degree of consistency. Accordingly, we chose a random-effects model. All included studies were cohort studies, and the data were two-category outcomes. Therefore, we extracted the total number of studies and the number of occurrences in the studies and calculated the ratio of the incidence of outcomes

TABLE 1 Characteristics of the studies that were included.

Study	Country	Study design	Sample size ^a	Male	Median age (IQR)(years)	Tumor type	Diagnostic method of COVID-19	Time ^b	Outcomes	NOS ^g
Hatic H, 2022 (39)	USA	Retrospective	121	70 (57.9%)	63.7#	No special	1,2 ^c	Within 365 days	Mortality and ICU	9
Isgrò M A,2021 (40)	Italy	Retrospective	885	568 (64.2%)	NA ^d	No special	2	Within 45 days	Infection	7
Luo J,2020 (41)	USA	Retrospective	69	33 (47.8%)	69.0 (31.0–91.0)	Lung cancer	1	Within a median of 45 days	Mortality and ICU	7
Jee J,2021 (42)	USA	Retrospective	608	NA	NA	No special	1	Within 90 days	Respiratory failure or mortality ^e	8
Lièvre A,2020 (43)	France	Retrospective	1289	795 (61.7%)	67.0 (19.0–100.0)	No special	1,2	within 90 days	Mortality	8
Mandala M,2021 (44)	Italy	Prospective	293	182 (62.1%)	NA	No special	1,2	within 90 days	Infection and mortality	7
Gonzalez-Cao M,2022 (<mark>45</mark>)	Spain	Prospective	150	90 (60.0%)	68.0 (6.0–95.0)	Melanoma	NA	NA	mortality	7
Lara O D,2022 (46)	USA	Retrospective	193	0 (0.0%)	65.0 (54.0–73.0)	Gynecologic cancer	1,2,4	within 90 days	Mortality and ICU	9
Bersanelli M,2020 (47)	Italy	Prospective	955	648 (67.9%)	69.5#	No special	1,4	within 90 days	Infection and mortality	7
Calles A,2020 (48)	Spain	Retrospective	242	137 (56.7%)	NA	Lung cancer	1	within 90 days	Infection and mortality	7
Fuentes Antrás J,2020 (49)	Spain	Prospective	73	41 (56.1%)	72.0 (59.0–82.0)	No special	1	NA	Mortality	7
Garassino M C,2020 (50)	Italy	Retrospective	200	141 (70.5%)	68.0 (61.8–75.0)	No special	1,3,4	Within a median of 7 days	Mortality	8
Nie L,2021 (51)	China	Retrospective	45	31 (68.9%)	66.0 (58.0–74.0)	Lung cancer	1	within 28 days	Mortality	7
Nichetti F, 2020 (52)	Italy	Retrospective	1081	491 (45.4%)	63.0 (19.0–91.0)	No special	1	within 54 days	Infection, mortality and ICU	7
Yarza R,2020 (53)	Spain	Prospective	63	34 (54.0%)	NA	No special	1,2,4	within 28 days	ARDS ^f	8

Frontiers in Immunology

Sun et al.

^aSample size means all patients in a study.

^bTime is the time of antitumor treatment before SARS-CoV-2 diagnosis.

^c1. RT-PCR; 2. Positive serology test, 3. suspected SARS-CoV-2 infection with symptoms; and 4. radiologic imaging.

^dNA means data not available.

^eThis outcome did not have separate data on mortality; therefore, it was divided into severity.

^fARDS means acute respiratory distress syndrome.

^gNOS means The Newcastle-Ottawa Scale.

Mean age.

between the intervention and control groups to obtain the RR (a more direct measure than OR). A forest plot was used to represent the metaanalysis results using Stata software.

Funnel plots were used to calculate publishing bias. Additionally, tests for Begg's rank correlation and Egger's linear regression were performed to demonstrate publication bias, with P <0.05 indicating significant publication bias. Additionally, sensitivity analysis was performed by eliminating one study at a time to determine whether the outcomes were affected by that particular study.

3 Results

3.1 Results of the electronic literature search

After 5,449 publications were retrieved from the electronic databases, 97 studies were selected for full-text screening based on

the eligibility requirements for titles and abstracts. Inconsistency in cancer type was the most common cause of exclusion in the full-text screening. In the end, 15 studies (39–53) were selected, having met all of the inclusion criteria. In Figure 1 we show the PRISMA diagram used to select the included research.

3.2 Study characteristics

The basic characteristics of the included studies are listed in Table 1. In total, 6,267 subjects who satisfied the inclusion criteria were included in the analysis. Retrospective cohort studies comprised the majority of the included studies, with five (citations 44, 45, 47, 49, 53) being prospectively designed. Among the 15 studies included, four (39, 41, 42, 46) were performed in the United States, five (40, 44, 47, 50, 52) in Italy, four (45, 48, 49, 53) in Spain, one (51) in China, and one (43) in France. All patients in the 15 studies had solid tumors: three of the studies (41, 48, 51) involved patients with lung cancer, one study (46) included



patients with gynecological tumors, and another study (45) included patients with melanoma; the remaining 10 studies were not specific as to cancer type.

In terms of the main outcomes, five studies (40, 44, 47, 48, 52) included the SARS-CoV-2 infection rate, and twelve (39, 41, 43–52) included the mortality rate. As a secondary result, the severity of SARS-CoV-2 was recorded in six of the studies (39, 41, 42, 46, 52, 53).

3.3 Impact of past ICIs exposure on SARS-CoV-2 infection in patients with solid cancer

Five studies analyzed the relationship between ICI treatment and the SARS-CoV-2 infection rate in patients with solid tumors; the pooled data showed no statistically significant differences (RR 1.04, 95% CI 0.57–1.88, z = 0.12; P = 0.905) (Figure 2).

We divided the study participants into two subgroups (retrospective and prospective) based on the study type. The RR of the retrospective subgroup was 1.01 (95% CI 0.36–2.91, z = 0.83, P = 0.407), compared with that of the prospective subgroup, 1.24 (95% CI 0.75–2.04, z = 0.02, P = 0.987). In the heterogeneity analysis, there was significant heterogeneity in the retrospective study subgroup (P = 0.038, $I^2 = 69.4\%$); however, the grouping factors were not the source of heterogeneity (P = 0.097, $I^2 = 49.0\%$). No significant heterogeneity was observed. As shown in Supplementary Material 3, meta-regression analyses were performed to identify potential sources of significant heterogeneity; there were no significant differences (P > 0.05).

The funnel plot of the SARS-CoV-2 infection rate showed publication bias (Supplementary Material 4); however, further investigation revealed that there was insufficient evidence of significant publication bias (Begg's test, P = 1.000; Egger's test, P = 0.873) (Supplementary Material 5). In addition, sensitivity

analyses that excluded each included study at a time revealed that the pooled RR values were not significantly influenced by the exclusion of any single study (Figure 2). Therefore, we conclude that the findings of this study are satisfactory.

3.4 Impact of past ICIs exposure on SARS-CoV-2 mortality in patients with solid cancer

Based on the meta-analysis, the pooled estimate RR of patients with past ICI exposure was 1.22 (95% CI 0.99–1.50, z = 1.90, P = 0.057) (Figure 3), however the RR in the ICI treatment group was not significantly different from the RR in the group that had not received ICI treatment.

A subgroup analysis (retrospective versus prospective) was performed. The average (pooled) RR of the retrospective subgroup was 1.17 (95% CI 0.94–1.45, z =1.43, P = 0.153), and the corresponding RR of the prospective subgroup was 1.73 (95% CI 0.92–3.26, z = 1.71, P = 0.088). In the heterogeneity analysis, there was no significant heterogeneity in either the retrospective (P = 0.764, $I^2 = 0.0\%$) or prospective study subgroups (P = 0.811, $I^2 = 0.0\%$), and the grouping factors were not the source of heterogeneity (P = 0.844, $I^2 = 0.0\%$). The findings of the metaregression are discussed in Supplementary Material 3. The investigated factors did not show heterogeneity (P > 0.05).

In examining the data from the Hatic H trial (2022) (39), we discovered that the anti-cancer treatment period before the diagnosis of SARS-CoV-2 was within 365 days; thus, the baseline was uneven compared to other studies. In constructing funnel plots of the included studies, we observed no publication bias, as shown in Supplementary Material 4. The results of Egger's linear regression and Begg's rank correlation tests are included in Supplementary Material 5 (Begg's test, P = 0.537; Egger's test, P = 0.881). Furthermore, the average effect size was unaffected by





sensitivity analyses that excluded one included study at a time (Figure 3). Therefore, the article in question was not deleted from our analysis.

3.5 Impact of past ICI exposure on SARS-CoV-2 severity in patients with solid cancers

Information on severe or critical diseases was reported in six of the studies, in which ICI was significantly linked with severe COVID-19 (pooled RR, 1.51; 95% CI, 1.09–2.10, z = 2.46, P = 0.014; Figure 4).

The results of the subgroup analysis showed that the pooled RR of the retrospective subgroup was 1.49 (95% CI: 1.06–2.11, z =2.27, P = 0.023), compared with 1.66 in the case of the prospective subgroup (95% CI: 0.60–4.61, z = 0.97, P = 0.334; this subgroup included only one study). In the heterogeneity analysis, there was no significant heterogeneity in either the retrospective study subgroup

 $(P = 0.648, I^2 = 0.0\%)$ or the prospective study subgroup (only one study), and the grouping factors were not the source of heterogeneity ($P = 0.774, I^2 = 0.0\%$). Meta-regression analysis (Supplementary Material 3) showed no heterogeneity (P > 0.05).

Neither the funnel plot (Supplementary Material 4) nor the tests for significant publication bias (Begg's test, P = 1.000; Egger's test, P = 0.930) (Supplementary Material 5) revealed any evidence of potential publication bias. In addition, regardless of excluding any particular study one at a time, the pooled RRs were not significantly influenced by the sensitivity analysis (Figure 4).

3.6 Cancer type

In terms of tumor types, we included cohort studies on solid tumors, covering lung cancer, gynecological tumors, melanoma, and other types of cancer; however, due to the lack of data in the included studies, we analyzed only the data of patients with lung cancer and observed that there was no significant correlation



between the treatment of ICIs and the infection rate (pooled RR, 1.55; 95% CI, 0.48–5.01, z = 0.73, P = 0.468; heterogeneity test, P = 0.039, $I^2 = 69.1\%$) and the mortality rate (pooled RR, 1.56; 95% CI, 0.63–3.84, z = 0.96, P = 0.338; heterogeneity test, P = 0.314, $I^2 = 13.7\%$). The specific data are provided in Supplementary Material 6.

In addition, owing to the inclusion of studies from the COVID pandemic period (from 2020 to 2022), we conducted a subgroup analysis of the years. The results showed that, in terms of infection, the pooled estimate RR was 1.04 (95% CI 0.57–1.88, z = 0.12, P = 0.905) (Supplementary Material 7), which was not statistically significant. The pooled RR of the 2021 subgroup was 0.80 (95% CI 0.31–2.06, z = 0.46, P = 0.648), compared with the corresponding value, 1.55, from the 2020 subgroup (95% CI 0.71–3.37, z = 1.11, P = 0.269). At the same time, in terms of mortality, the pooled estimate had no statistically significant differences (RR 1.22, 95% CI 0.99–1.50, z = 1.90, P = 0.057). The pooled RR of the 2022 subgroup was 1.29 (95% CI 0.69–2.44, z = 0.80, P = 0.424); that of the 2021 subgroup was 0.78 (95% CI 0.23–2.60, z = 0.41, P = 0.682); and 1.23 in the 2020 subgroup (95% CI 0.99–1.53, z = 1.84, P = 0.065).

4 Discussion

The outbreak of SARS-CoV-2 has disrupted many aspects of human life, especially health care. A meta-analysis by Muka et al. (54) confirmed that the pandemic had a significant impact on cancer care, including delays in screening, diagnosis, and treatment. In the short term, these results suggest that interventions are needed to mitigate the negative impacts of infectious disease epidemics on cancer care. In the long run, they demonstrated the importance of rigorous systematic reviews in guiding decision-making. To the best of our knowledge, this systematic review and meta-analysis is the first to assess the association between ICIs and SARS-CoV-2 infection and prognosis in patients with solid cancers. The analysis was performed on pooled data from 15 cohort studies. We used a random-effects model to fit the actual sampling distribution as much as possible and extended the conclusion to a wide range of scenarios.

In addition to cancer immunosurveillance, the immune system plays a crucial role in protecting against pathogens, bacteria, viruses, and fungi. T cells are continually subjected to antigens, including recurrent viral infections, which contribute to T cell exhaustion. This implies that ICIs may increase CD8+ T cell proliferative capabilities by obstructing CTLA4 and PD-1/PD-L1-mediated signaling pathways, theoretically lowering the viral load (55). PD-1-targeted treatment inhibits tumor growth and lowers viral load in various cancer and persistent infection mouse models (56). The same observations have been made in the case of non-human primates, Yatim et al. (57) provided evidence that ICI during COVID-19 enhanced T-cell immunity without exacerbating inflammation. Additionally, patients with cancer receiving ICIs can recover their immunocompetence after contracting human immunodeficiency virus (HIV), hepatitis B, or hepatitis C viruses, indicating that they may be significantly more immunocompetent than other patients with cancer receiving chemotherapy or radiation therapy (58). Consequently, several therapeutic trials, including the use of immune checkpoint blockade in treating recurrent viral infections, have been designed and administered to patients with cancer (59).

The main outcomes of our study support the judicious use of ICIs in treating patients with cancer who have been previously exposed to SARS-CoV-2. Our main outcomes showed that prior exposure of patients with solid cancers to ICIs was not linked to a higher risk of SARS-CoV-2 infection or mortality (P > 0.05). However, the analysis showed that (based on six of the selected studies), the use of ICIs may aggravate the condition of SARS-CoV-2 in patients with solid cancer and increase the risk of adverse clinical outcomes (P < 0.05). Considering these results together, the conclusion is that practitioners need to carefully evaluate the benefits and drawbacks of the use of ICIs in patients with solid tumors, considering their history of SARS-CoV-2 infection, and proceed with caution.

To address potential heterogeneity in the selected study population, we conducted a subgroup analysis of the study types and observed that the differences between prospective and retrospective studies were not a source of heterogeneity. In addition, considering the possible impact of the different prevalence rates of SARS-CoV-2 from 2020 to 2022 on the results, we conducted an analysis of subgroups by year and observed no significant correlation.

Among all solid tumors, lung cancer is the most dangerous to patients who had been infected with SARS-CoV-2 prior to their cancer treatment (60). Patients with lung tumors tend to have relatively poor physical status. Smoking habits, and multiple comorbid diseases often present with comorbid lung disease and poor lung function (61) and are predisposed to respiratory infections. Therefore, we constructed a forest plot for a small portion of the study cohort, those patients with lung cancer. Our results showed no significant correlation between ICIs treatment and the SARS-CoV-2 infection and mortality rates of patients with lung cancer. SARS-CoV-2 does not tend to be more severe in lung cancer cases. In future studies, a greater range of literature with adequate and transparent reporting of tumor types should be included to confirm the applicability of our findings.

Our study observed that ICIs were not associated with a higher SARS-CoV-2 infection rate or mortality in patients with solid tumors, which is inconsistent with previous research results (62). We note that previous studies have combined hematological malignancies as one of the possible reasons for the difference in results. In contrast to most solid tumors, hematological malignancies are highly susceptible to interactions with immune cells and are responsive to ongoing systemic inflammatory reactions (63). Moreover, whereas patients with solid tumors appear to no longer be at risk of SARS-CoV-2-associated immune dysregulation compared to the general population, patients with hematological cancer display complicated immunological consequences of SARS-CoV-2 exposure (64) and tend to have poor prognosis (65, 66). For this reason, we analyzed data from solid tumor cases, rather than other forms of cancer.

Our findings are supported by those of a few previous studies. In a systematic analysis, Lazarus et al. (29) included 11 trials with a total of 2,826 patients with cancer infected with COVID-19. The authors observed that moderate-to-high-quality evidence for ICIs was not associated with an increased mortality risk. According to Liu et al. (30), who analyzed data from 5,121 patients with cancer and COVID-19, there was no discernible difference in mortality between groups of patients undergoing anti-tumor therapy (including immunotherapy) and those who were not.

This systematic review has several strengths. To the best of our knowledge, this is the first comprehensive analysis of SARS-CoV-2 infection and prognosis in patients with solid tumors receiving ICIs. Second, it incorporates a sizable number of studies not covered in earlier evaluations, making it the most thorough and reliable body of information available to date. Third, minimal or non-existent publication bias and inter-study increases our confidence in the findings.

Our analysis has several limitations. First, although our dataset included different types of cancer, we chose to separate the data associated with cases of lung cancer (and produced a forest plot from this data). The results suggest that there is no significant correlation between ICI treatment and the SARS-CoV-2 infection and mortality rates in patients with lung cancer. However, because of the small sample size and relative paucity of studies, we cannot guarantee the accuracy of the results. Subgroup analysis based on cancer type was not performed because other tumor types could not be separated from the valid data. In terms of differences in cancer staging, we did not find the corresponding outcome indicator data in the literature. Moreover, there will be more research on a certain type or stage of cancer in this field in the future. Second, we set the control group as patients receiving other antitumor treatments (other than ICI). Differences exist in the modes of action between the antitumor treatments, which may have adversely affected the results. Currently, the literature is not sufficient to allow a robust comparison of a particular anti-tumor treatment with ICIs. In the future, we will actively conduct more narrowly defined grouping studies while expanding the dataset to include a greater number of cases that meet the inclusion criteria.

5 Conclusion

This is the first meta-analysis in which the possible association between ICI treatment and SARS-CoV-2 infection and prognosis was evaluated using retrospective data from patients with solid cancers. Our main findings are that the prior exposure of patients with solid cancer to ICIs was not linked to a higher risk of SARS-CoV-2 infection or mortality. However, in the secondary outcomes, ICI use was significantly associated with a greater probability of severe/critical disease in individuals with solid tumors before SARS-CoV-2 diagnosis. Therefore, we believe that in the current environment, the use of ICIs is not categorically disadvantageous to patients with solid tumors, but caution must be exercised.

Data availability statement

The original contributions presented in the study are included in the article/Supplementary material. Further inquiries can be directed to the corresponding author.

Author contributions

LS: Conceptualization, Formal analysis, Investigation, Software, Validation, Writing – original draft. FZ: Conceptualization, Formal analysis, Investigation, Software, Validation, Writing – review & editing. YX: Conceptualization, Validation, Writing – review & editing. SC: Project administration, Validation, Writing – review & editing. QS: Conceptualization, Supervision, Writing – review & editing.

Funding

The author(s) declare that no financial support was received for the research, authorship, and/or publication of this article.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Supplementary material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fimmu.2024.1259112/full#supplementary-material

References

 Calabrò L, Peters S, Soria JC, Di Giacomo AM, Barlesi F, Covre A, et al. Challenges in lung cancer therapy during the covid-19 pandemic. *Lancet Respir Med.* (2020) 8:542–4. doi: 10.1016/S2213–2600(20)30170–3

2. Curigliano G. Cancer patients and risk of mortality for covid-19. Cancer Cell. (2020) 38:161-3. doi: 10.1016/j.ccell.2020.07.006

3. Fillmore NR, La J, Szalat RE, Tuck DP, Yildirim C, Do NV, et al. Prevalence and outcome of covid-19 infection in cancer patients: A national veterans affairs study. *J Natl Cancer Inst.* (2021) 113:691–8. doi: 10.1093/jnci/djaa159

4. Rüthrich MM, Giessen-Jung C, Borgmann S, Classen AY, Dolff S, Grüner B, et al. Covid-19 in cancer patients: clinical characteristics and outcome-an analysis of the leoss registry. *Ann Hematol.* (2021) 100:383–93. doi: 10.1007/s00277-020-04328-4

5. Ribas A, Wolchok JD. Cancer immunotherapy using checkpoint blockade. Science. (2018) 359:1350-5. doi: 10.1126/science.aar4060

6. NCCN. Prevention and treatment of cancer-related infections (2022). Available online at: https://www.nccn.org/guidelines/guidelines-detail?category=3&id=1457.

7. Giesen N, Sprute R, Rüthrich M, Khodamoradi Y, Mellinghoff SC, Beutel G, et al. Evidence-based management of covid-19 in cancer patients: guideline by the infectious diseases working party (Agiho) of the german society for hematology and medical oncology (Dgho). *Eur J Cancer*. (2020) 140:86–104. doi: 10.1016/j.ejca.2020.09.009

8. NICE. Covid-19 rapid guideline: delivery of systemic anticancer treatments (2022). Available online at: https://www.nice.org.uk/guidance/ng161.

9. Hanna TP, King WD, Thibodeau S, Jalink M, Paulin GA, Harvey-Jones E, et al. Mortality due to cancer treatment delay: systematic review and meta-analysis. *Bmj.* (2020) 371:m4087. doi: 10.1136/bmj.m4087

10. Bardet A, Fraslin AM, Marghadi J, Borget I, Faron M, Honoré C, et al. Impact of covid-19 on healthcare organization and cancer outcomes. *Eur J Cancer*. (2021) 153:123–32. doi: 10.1016/j.ejca.2021.05.012

11. Malagón T, Yong JHE, Tope P, Miller WH Jr., Franco EL. Predicted long-term impact of covid-19 pandemic-related care delays on cancer mortality in Canada. *Int J Cancer*. (2022) 150:1244–54. doi: 10.1002/ijc.33884

12. Yekedüz E, Utkan G, Ürün Y. A systematic review and meta-analysis: the effect of active cancer treatment on severity of covid-19. *Eur J Cancer (Oxford England: 1990).* (2020) 141:92–104. doi: 10.1016/j.ejca.2020.09.028

13. Pezeshki PS, Rezaei N. Immune checkpoint inhibition in covid-19: risks and benefits. *Expert Opin Biol Ther.* (2021) 21:1173–9. doi: 10.1080/14712598.2021.1887131

14. Garassino MC, Ribas A. At the crossroads: covid-19 and immune-checkpoint blockade for cancer. *Cancer Immunol Res.* (2021) 9:261–4. doi: 10.1158/2326-6066.CIR-21-0008

15. Cai H, Liu G, Zhong J, Zheng K, Xiao H, Li C, et al. Immune checkpoints in viral infections. *Viruses*. (2020) 12:1051. doi: 10.3390/v12091051

16. Okuhira H, Yamamoto Y, Inaba Y, Kunimoto K, Mikita N, Ikeda T, et al. Prognostic factors of daily blood examination for advanced melanoma patients treated with nivolumab. *Biosci Trends.* (2018) 12:412–8. doi: 10.5582/bst.2018.01158

17. Schmidt ME, Varga SM. The cd8 T cell response to respiratory virus infections. *Front Immunol.* (2018) 9:678. doi: 10.3389/fimmu.2018.00678

18. Lineburg KE, Grant EJ, Swaminathan S, Chatzileontiadou DSM, Szeto C, Sloane H, et al. Cd8(+) T cells specific for an immunodominant sars-cov-2 nucleocapsid epitope cross-react with selective seasonal coronaviruses. *Immunity.* (2021) 54:1055–65.e5. doi: 10.1016/j.immuni.2021.04.006

19. Gao M, Liu Y, Guo M, Wang Q, Wang Y, Fan J, et al. Regulatory cd4(+) and cd8 (+) T cells are negatively correlated with cd4(+) /cd8(+) T cell ratios in patients acutely infected with sars-cov-2. *J Leukoc Biol.* (2021) 109:91–7. doi: 10.1002/JLB.5COVA0720-421RR

20. Li M, Guo W, Dong Y, Wang X, Dai D, Liu X, et al. Elevated exhaustion levels of nk and cd8(+) T cells as indicators for progression and prognosis of covid-19 disease. *Front Immunol.* (2020) 11:580237. doi: 10.3389/fimmu.2020.580237

21. Anft M, Paniskaki K, Blazquez-Navarro A, Doevelaar A, Seibert FS, Hölzer B, et al. Covid-19-induced ards is associated with decreased frequency of activated memory/effector T cells expressing cd11a(+). *Mol Ther*. (2020) 28:2691–702. doi: 10.1016/j.ymthe.2020.10.001

22. Wykes MN, Lewin SR. Immune checkpoint blockade in infectious diseases. Nat Rev Immunol. (2018) 18:91–104. doi: 10.1038/nri.2017.112

23. Hatic H, Sampat D, Goyal G. Immune checkpoint inhibitors in lymphoma: challenges and opportunities. *Ann Transl Med.* (2021) 9:1037. doi: 10.21037/atm-20-6833

24. Greaves P, Gribben JG. The role of B7 family molecules in hematologic Malignancy. Blood. (2013) 121:734-44. doi: 10.1182/blood-2012-10-385591

25. Xie W, Medeiros LJ, Li S, Yin CC, Khoury JD, Xu J. Pd-1/pd-L1 pathway and its blockade in patients with classic hodgkin lymphoma and non-hodgkin large-cell lymphomas. *Curr Hematol Malig Rep.* (2020) 15:372–81. doi: 10.1007/s11899-020-00589-y

26. Kuzume A, Chi S, Yamauchi N, Minami Y. Immune-checkpoint blockade therapy in lymphoma. *Int J Mol Sci.* (2020) 21:5456. doi: 10.3390/ijms21155456

27. Malas S, Harrasser M, Lacy KE, Karagiannis SN. Antibody therapies for melanoma: new and emerging opportunities to activate immunity (Review). Oncol Rep. (2014) 32:875–86. doi: 10.3892/or.2014.3275

28. Rutigliano JA, Sharma S, Morris MY, Oguin TH 3rd, McClaren JL, Doherty PC, et al. Highly pathological influenza a virus infection is associated with augmented expression of pd-1 by functionally compromised virus-specific cd8+ T cells. *J Virol.* (2014) 88:1636–51. doi: 10.1128/jvi.02851–13

29. Lazarus G, Budiman RA, Rinaldi I. Does immune checkpoint inhibitor increase the risks of poor outcomes in covid-19-infected cancer patients? A systematic review and meta-analysis. *Cancer Immunol. Immunother*. (2022) 71:373–86. doi: 10.1007/s00262–021-02990–9

30. Liu H, Yang D, Chen X, Sun Z, Zou Y, Chen C, et al. The effect of anticancer treatment on cancer patients with covid-19: A systematic review and meta-analysis. *Cancer Med.* (2021) 10:1043–56. doi: 10.1002/cam4.3692

31. Page MJ, Moher D, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, et al. Prisma 2020 explanation and elaboration: updated guidance and exemplars for reporting systematic reviews. *Bmj*. (2021) 372:n160. doi: 10.1136/bmj.n160

32. WHO. Therapeutics and covid-19: living guideline (2023). Available online at: https://www.who.int/publications/i/item/WHO-2019-nCoV-therapeutics-2023.1.

33. WHO. Corticosteroids for covid-19 (2020). Available online at: https://www.who.int/publications/i/item/WHO-2019-nCoV-Corticosteroids-2020.1.

34. Wells G, Shea B, O'Connell D, Peterson J, Welch V, Losos M, et al. The newcastle-ottawa scale (Nos) for assessing the quality of nonrandomized studies in meta-analyses (2009). Available online at: https://www.ohri.ca/programs/clinical_epidemiology/oxford.asp.

35. Stang A. Critical evaluation of the newcastle-ottawa scale for the assessment of the quality of nonrandomized studies in meta-analyses. *Eur J Epidemiol*. (2010) 25:603–5. doi: 10.1007/s10654-010-9491-z

36. Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. *Bmj.* (2003) 327:557–60. doi: 10.1136/bmj.327.7414.557

37. Higgins JP, Thompson SG. Quantifying heterogeneity in a meta-analysis. *Stat Med.* (2002) 21:1539–58. doi: 10.1002/sim.1186

38. Borenstein M, Hedges LV, Higgins JP, Rothstein HR. A basic introduction to fixed-effect and random-effects models for meta-analysis. *Res Synth Methods*. (2010) 1:97–111. doi: 10.1002/jrsm.12

39. Hatic H, Hearld KR, Das D, Deshane J. Clinical outcomes in covid-19 patients treated with immunotherapy. *Cancers (Basel).* (2022) 14:5954. doi: 10.3390/cancers14235954

40. Isgrò MA, Vitale MG, Celentano E, Nocerino F, Porciello G, Curvietto M, et al. Immunotherapy may protect cancer patients from sars-cov-2 infection: A single-center retrospective analysis. J Trans Med. (2021) 19:132. doi: 10.1186/s12967-021-02798-2

41. Luo J, Rizvi H, Egger JV, Preeshagul IR, Wolchok JD, Hellmann MD. Impact of pd-1 blockade on severity of covid-19 in patients with lung cancers. *Cancer Discovery*. (2020) 10:1121-8. doi: 10.1158/2159-8290.CD-20-0596

42. Jee J, Stonestrom AJ, Devlin S, Nguyentran T, Wills B, Narendra V, et al. Oncologic immunomodulatory agents in patients with cancer and covid-19. *Sci Rep.* (2021) 11:4814. doi: 10.1038/s41598-021-84137-5

43. Lièvre A, Turpin A, Ray-Coquard IL, Le Malicot K, Thariat J, Ahle G, et al. Risk factors for coronavirus disease 2019 (Covid-19) severity and mortality among solid cancer patients and impact of the disease on anticancer treatment: A french nationwide cohort study (Gco-002 cacovid-19). *Eur J Cancer.* (2020) 141:62–81. doi: 10.1016/j.ejca.2020.09.035

44. Mandala M, Lorigan P, De Luca M, Bianchetti A, Merelli B, Bettini AC, et al. Sars-cov-2 infection and adverse events in patients with cancer receiving immune checkpoint inhibitors: an observational prospective study. *J immunother. Cancer.* (2021) 9:e001694. doi: 10.1136/jitc-2020-001694

45. Gonzalez-Cao M, Puertolas T, Martinez-Vila C, Carrera C, Maldonado Seral C, Rodríguez-Jiménez P, et al. Sars-cov-2 infection in patients with melanoma: results of the spanish melanoma group registry. *Clin Transl Oncol.* (2022) 25:768–75. doi: 10.1007/s12094-022-02985-7

46. Lara OD, Smith M, Wang Y, O'Cearbhaill RE, Blank SV, Kolev V, et al. Covid-19 outcomes of patients with gynecologic cancer in new york city: an updated analysis from the initial surge of the pandemic. *Gynecol Oncol.* (2022) 164:304–10. doi: 10.1016/j.ygyno.2021.12.004

47. Bersanelli M, Giannarelli D, De Giorgi U, Pignata S, Di Maio M, Verzoni E, et al. Symptomatic covid-19 in advanced-cancer patients treated with immune-checkpoint inhibitors: prospective analysis from a multicenter observational trial by ficog. *Ther Adv Med Oncol.* (2020) 12:1758835920968463. doi: 10.1177/1758835920968463

48. Calles A, Aparicio MI, Alva M, Bringas M, Gutierrez N, Soto J, et al. Outcomes of covid-19 in patients with lung cancer treated in a tertiary hospital in madrid. *Front Oncol.* (2020) 10:1777. doi: 10.3389/fonc.2020.01777

49. Fuentes-Antrás J, Manzano A, Marquina G, Paz M, Aguado C, Granja M, et al. A snapshot of covid-19 infection in patients with solid tumors. *Int J Cancer*. (2020) 148:2389–97. doi: 10.1002/ijc.33420

50. Garassino MC, Whisenant JG, Huang LC, Trama A, Torri V, Agustoni F, et al. Covid-19 in patients with thoracic Malignancies (Teravolt): first results of an international, registrybased, cohort study. *Lancet Oncol.* (2020) 21:914–22. doi: 10.1016/s1470-2045(20)30314–4

51. Nie L, Dai K, Wu J, Zhou X, Hu J, Zhang C, et al. Clinical characteristics and risk factors for in-hospital mortality of lung cancer patients with covid-19: A multicenter, retrospective, cohort study. *Thorac Cancer*. (2021) 12:57–65. doi: 10.1111/1759–7714.13710

52. Nichetti F, Bini M, Ambrosini M, Ottini A, Rametta A, Leporati R, et al. Covid-19 risk for patients undergoing anticancer treatment at the outpatient clinic of the national cancer institute of milan: the covint study. *ESMO Open*. (2020) 5:e000883. doi: 10.1136/esmoopen-2020-000883

53. Yarza R, Bover M, Paredes D, López-López F, Jara-Casas D, Castelo-Loureiro A, et al. Sars-cov-2 infection in cancer patients undergoing active treatment: analysis of clinical features and predictive factors for severe respiratory failure and death. *Eur J Cancer (Oxford England: 1990).* (2020) 135:242–50. doi: 10.1016/j.ejca.2020.06.001

54. Muka T, Li JJX, Farahani SJ, Ioannidis JPA. An umbrella review of systematic reviews on the impact of the covid-19 pandemic on cancer prevention and management, and patient needs. *Elife*. (2023) 12:e85679. doi: 10.7554/eLife.85679

55. Gatto L, Franceschi E, Nunno VD, Brandes AA. Potential protective and therapeutic role of immune checkpoint inhibitors against viral infections and covid-19. *Immunotherapy*. (2020) 12:1111-4. doi: 10.2217/imt-2020-0109

56. Im SJ, Hashimoto M, Gerner MY, Lee J, Kissick HT, Burger MC, et al. Defining cd8+ T cells that provide the proliferative burst after pd-1 therapy. *Nature*. (2016) 537:417-21. doi: 10.1038/nature19330

57. Yatim N, Boussier J, Tetu P, Smith N, Bruel T, Charbit B, et al. Immune checkpoint inhibitors increase T cell immunity during sars-cov-2 infection. *Sci Adv.* (2021) 7:eabg4081. doi: 10.1126/sciadv.abg4081

58. Shah NJ, Al-Shbool G, Blackburn M, Cook M, Belouali A, Liu SV, et al. Safety and efficacy of immune checkpoint inhibitors (Icis) in cancer patients with hiv,

hepatitis B, or hepatitis C viral infection. J Immunother Cancer. (2019) 7:353. doi: 10.1186/s40425-019-0771-1

59. Dyck L, Mills KHG. Immune checkpoints and their inhibition in cancer and infectious diseases. *Eur J Immunol.* (2017) 47:765–79. doi: 10.1002/eji.201646875

60. Dai M, Liu D, Liu M, Zhou F, Li G, Chen Z, et al. Patients with cancer appear more vulnerable to sars-cov-2: A multicenter study during the covid-19 outbreak. *Cancer Discovery.* (2020) 10:783. doi: 10.1158/2159-8290.CD-20-0422

61. Xu Y, Liu H, Hu K, Wang M. Clinical recommendations on lung cancer management during the covid-19 pandemic. *Thorac Cancer.* (2020) 11:2067–74. doi: 10.1111/1759-7714.13498

62. Robilotti EV, Babady NE, Mead PA, Rolling T, Perez-Johnston R, Bernardes M, et al. Determinants of covid-19 disease severity in patients with cancer. *Nat Med.* (2020) 26:1218–+. doi: 10.1038/s41591–020-0979–0

63. Latif MB, Shukla S, Del Rio Estrada PM, Ribeiro SP, Sekaly RP, Sharma AA. Immune mechanisms in cancer patients that lead to poor outcomes of sars-cov-2 infection. *Transl Res.* (2022) 241:83–95. doi: 10.1016/j.trsl.2021.12.001

64. Abdul-Jawad S, Baù L, Alaguthurai T, del Molino del Barrio I, Laing AG, Hayday TS, et al. Acute immune signatures and their legacies in severe acute respiratory syndrome coronavirus-2 infected cancer patients. *Cancer Cell.* (2021) 39:257–75.e6. doi: 10.1016/j.ccell.2021.01.001

65. Lee LYW, Cazier JB, Angelis V, Arnold R, Bisht V, Campton NA, et al. Covid-19 mortality in patients with cancer on chemotherapy or other anticancer treatments: A prospective cohort study. *Lancet.* (2020) 395:1919–26. doi: 10.1016/S0140-6736(20) 31173–9

66. Passamonti F, Cattaneo C, Arcaini L, Bruna R, Cavo M, Merli F, et al. Clinical characteristics and risk factors associated with covid-19 severity in patients with hematological Malignancies in Italy: A retrospective, multicentre, cohort study. *Lancet Haematol.* (2020) 7:e737–e45. doi: 10.1016/s2352–3026(20)30251–9